

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Brücker, Herbert; Jahn, Elke J.

Conference Paper Migration and Wage-Setting: Reassessing the Labor Market Effects of Migration

Beiträge zur Jahrestagung des Vereins für Socialpolitik 2010: Ökonomie der Familie - Session: Determinants of Unemployment, No. D6-V1

Provided in Cooperation with:

Verein für Socialpolitik / German Economic Association

Suggested Citation: Brücker, Herbert; Jahn, Elke J. (2010) : Migration and Wage-Setting: Reassessing the Labor Market Effects of Migration, Beiträge zur Jahrestagung des Vereins für Socialpolitik 2010: Ökonomie der Familie - Session: Determinants of Unemployment, No. D6-V1, Verein für Socialpolitik, Frankfurt a. M.

This Version is available at: https://hdl.handle.net/10419/37359

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

Migration and Wage-Setting: Reassessing the Labor Market Effects of Migration

Herbert Brücker^{*} and Elke J. Jahn[‡]

This version: January 2010

Abstract

This paper employs a wage-setting approach to analyze the labor market effects of immigration into Germany from 1980 to 2004. This enables us to consider labor market rigidities, which are prevalent in Europe. We find that the elasticity of the wage-setting curve is particularly high for young workers. Moreover, natives and foreigners are imperfect substitutes. The wage and employment effects of immigration depend on the skill structure of the immigrant workforce. Since the foreign labor supply shift mainly affected the high-skilled labor market segment, the four percent increase of the workforce through immigration did neither increase aggregate nor foreign unemployment.

Keywords: immigration, wages, unemployment, Germany, panel data. *JEL code*: F22, J31, J61.

^{*}University of Bamberg, IAB Nürnberg and IZA Bonn, Herbert.Bruecker@iab.de.

[†]Århus School of Business, IAB Nürnberg and IZA Bonn, elja@asb.dk.

[‡]The authors are grateful to Andreas Hauptmann who programmed the simulation model, Thomas Bttner, who imputed the wages, Timo Baas, Uwe Blien, Jennifer Hunt, Concetta Mendolicchio, Dario Pozzoli, Michael Rosholm, Claus Schnabel, Jrgen Wiemers and to the referees. We gratefully acknowledge financial support from the EU "Transnationality of Migrants" project.

1 Introduction

High and increasing immigration rates in the US and Europe have fanned fears that migrants reduce wages and limit the employment opportunities of the native labor force. In continental European countries – where labor market rigidities are prevalent and unemployment is already high and persistent – concerns that immigration will exacerbate unemployment are particularly widespread. Moreover, disproportionately high unemployment rates among the foreign workforce have contributed to popular perceptions that further immigration will create an unsustainable fiscal burden on European welfare states. If immigration does indeed have adverse effects on the labor market, continental European governments may come under further pressure to tighten their immigration policies.

In this paper we present a new approach to measuring the wage and employment effects of migration under the conditions of imperfect labor markets. With this approach, we are able to address the effects of migration on unemployment in greater depth than earlier studies have done. Our approach is based on a wage-setting framework in which we assume that wages tend to decline with increasing unemployment under given price expectations, albeit imperfectly. Collective bargaining and efficiency wage models suggest that the elasticity between wages and the unemployment rate varies between groups in the labor market (Layard et al. 2005). In our empirical analysis we therefore estimate wage-setting curves that, together with the elasticities of labor demand, determine the wage and employment effects of immigration in different segments of the labor market. This enables us to take into account labor market rigidities, which exercise a particularly strong impact in the European context, affecting different types of labor in different ways.

Our empirical analysis focuses on Germany, the third most popular destination for migrants in the world after the US and Russia (Freeman 2006). Following the 1989 collapse of the Communist regimes of Central and Eastern Europe, the net immigration rate in Western Germany climbed from about zero in the early 1980s to about six per thousand at the beginning of the 1990s, compared to three per thousand in the fifteen member states of the European Union (EU-15) and four per thousand in the US (World Bank 2007). However, since the beginning of this millennium, the net immigration rate has dropped to less than three per thousand in the wake of Germany's economic downturn.

Germany is also an interesting case because it has been suffering from high and persistent unemployment since the first oil price shock in 1973. The unemployment rate has increased further since German unification in 1990. Moreover, due to Germany's extensive labor market regulations, the labor market is considered to be highly rigid (OECD 1994), and collective bargaining still plays an import role in wage-setting. About 85 percent of employees are directly or indirectly covered by collective agreements negotiated mainly at the industry level (Ellguth/Kohaut 2007). However, efficiency wages play an important role as well. About 46 percent of the firms bound by collective agreements pay wages above the levels stipulated in the agreements (Jung/Schnabel 2009).

While our approach derives the labor market effects of immigration from an equilibrium framework, the overwhelming majority of the empirical literature estimates reduced-form equations that relate wages or employment variables to the immigrant share in specific geographic areas or industries (Card 1990, Hunt 1992, Pischke/Velling 1997, Dustmann et al. 2005). These studies find only very small wage and employment effects, if any at all (Longhi et al. 2006, 2008). The spatial correlation approach has been criticized, however, for yielding spurious results if immigrants are not randomly distributed across local labor markets (or industries) or if other factors equilibrate labor market conditions across geographical areas (Borjas 2003). Some recent studies have therefore used the variance in immigrant shares across education and experience groups at the national level, assuming that the allocation of immigrants across skill and experience groups is exogenous (Borjas 2003, Aydemir/Borjas 2006). Nevertheless, this literature estimates only partial correlations between wage or employment variables on the one hand and the immigrant share on the other, and does not consider the interaction between wages and employment and the cross-effects of labor supply shifts in different segments of the labor market.

Another strand of the empirical literature derives the wage effects of migration from an aggregate production function approach (Grossmann 1992, Borjas 2003, Ottaviano/Peri 2006, 2008, Manacorda et al. 2006, Dustmann et al. 2008). This general equilibrium approach takes cross-effects of labor supply shifts on wages in different segments of the labor market into account, but it relies on the assumptions of perfect competition and clearing labor markets and thus cannot address the effects of migration on wages and employment if labor markets are imperfect.

Closer to our approach are some recent studies that analyze the effects of immigration on the Phillips curve. Binyamini/Razin (2008) and Engler (2007) show that migration can alter the elasticity of labor demand and supply, inducing a flatter Phillips curve. Similarly, Bentolila et al. (2008) find that immigration has reduced the wage elasticity of labor supply and the wage-markup in Spain, which in turn has weakened the trade-off between inflation and unemployment there. The present paper differs from these contributions in that it focuses on the long-run equilibrium relationship between wages and unemployment and does not address the impact of immigration on the inflation-unemployment trade-off. Moreover, we consider disaggregated labor supply shifts in different education and experience segments of the labor market rather than aggregate labor supply shocks.

Three recent papers that address the wage and employment effects of immigration in Germany are related to our study: Glitz (2006) examines the labor market effects of the large-scale immigration of ethnic Germans (so-called "Sptaussiedler") based on the spatial correlation approach. Using the IAB employment sample, an earlier version of the data set we use for our empirical analysis, he treats the immigration of ethnic Germans as a natural experiment and finds no detrimental impact on wages but strong displacement effects. Although the partial correlation approach in that paper differs from the equilibrium framework used here, his findings indicate that wage rigidities are prevalent in Germany.

In two other recent contributions, D'Amuri et al. (2008) and Felbermayr et al. (2008), apply a general equilibrium framework to the analysis of the labor market effects of immigration in Germany. Although both papers highlight the importance of unemployment and wage rigidities, their approach differs from ours. Similarly to Borjas (2003) and Ottaviano/Peri (2006), both papers derive the wage effects from a nested production function framework. Their identification strategy relies implicitly on the assumption that labor markets clear. Following the standard procedure in the literature, D'Amuri et al. (2008) estimate the employment effects separately from the wage effects in reduced-form equations, while our paper determines the wage and employment effects simultaneously in an equilibrium framework with imperfect labor markets. There are further important differences between these papers and ours in the identification strategy and data sources. Felbermayr et al. (2008) base their analysis on the German Socio-Economic Panel (GSOEP), which suffers from attenuation bias since the foreigner shares in specific education and experience cells are very small there. D'Amuri et al. (2008) use an earlier version of the IABS but impute the substantial immigration of ethnic Germans during the 1990s, while we identify them directly using information on their participation in labor market programs. We imputed the education variable using a procedure especially developed for the IABS that improves its quality considerably. This point is particularly relevant in case of foreign workers. Furthermore, D'Amuri et al. (2008) only estimate the elasticity of substitution between natives and foreigners and take the elasticities of substitution across education and experience cells from the US literature, while we estimate the full set of elasticities.

The remainder of this paper is organized as follows. Section 2 outlines our theoretical framework. Section 3 describes the data set. Section 4 presents

the identification strategy and the estimation results for the elasticities of the wage-setting curves and the parameters of the production function. Section 5 simulates the employment and wage impact of immigration on the German labor market. Finally, Section 6 concludes.

2 Theoretical background

Building on Boeri/Brcker (2005) and Levine (1999) we apply a wage-setting framework to analyze the wage and employment effects of immigration (Layard/Nickell 1986, Layard et al. 2005). Our model replaces the conventional labor supply curve with a wage-setting function, which is consistent with right-to-manage models of collective bargaining (Nickell/Andrews 1983) and efficiency wage theories derived from turnover cost (Salop 1979) or shirking (Shapiro/Stiglitz 1984) models. We do not present an explicit collective bargaining or efficiency wage model here, since both types of models are relevant in the German context. Depending on the bargaining power of workers or human capital characteristics relevant to efficiency wage considerations, the slope of the wage-setting curve may vary in the different segments of the labor market. We therefore follow a suggestion by Card (1995) and allow the elasticity of the wage-setting curve to differ across education and experience groups of the labor market.

2.1 The wage-setting model

Suppose an economy where identical firms produce varieties of a differentiated good under monopolistic competition using different types of labor, L_{ℓ} , $\ell = 1, \ldots, n$, and physical capital. Production involves some fixed setup costs, but thereafter each firm produces output with constant returns to scale. Hence, production of a representative firm is given by

$$Y^{i} = F\left(\mathbf{L}^{i}, K^{i}\right),\tag{1}$$

where Y^i denotes a variety of the output good, L^i the vector of labor inputs, K^i physical capital and the superscript *i* the firm index. The production technology $F(\cdot)$ is increasing, concave, twice continuously differentiable in all inputs and homogeneous of degree one.

Let \overline{N}_{ℓ} , $\ell = 1, \ldots, n$ be the pre-migration labor force in each cell of the labor market. The post-migration labor force is then given by

$$N_{\ell} = \overline{N}_{\ell} + \gamma_{\ell} M, \qquad \sum_{\ell} \gamma_{\ell} = 1, \qquad (2)$$

where M is the total labor influx and γ_{ℓ} is the share of workers of type ℓ in the total immigrant inflow.

Wages and the demand for labor are determined sequentially. In the first stage, wages are fixed, whereas in the second stage, given the agreed wages, firms set prices and hire workers up to a level where profits are maximized. Suppose that each firm faces a constant elasticity of demand $\eta > 1$. Profit maximization implies that the wage rate equal

$$w_{\ell}^{i} = \nu^{-1} P^{i} Y_{L_{\ell}}^{i}, \qquad \forall \ell,$$

where the mark-up, ν , is $(1 - 1/\eta)^{-1}$, P^i the product price of variety *i* of the output good, and $Y^i_{L_{\ell}}$ the marginal product of labor.

With identical firms we can move to the level of the aggregate economy by writing $w_{\ell}^{i} = w_{\ell}$, $Y_{L_{\ell}}^{i} = Y_{L_{\ell}}$, and $P^{i} = P = 1$, where we have normalized the price level to one. The real wage is then given by

$$w_{\ell} = \nu^{-1} Y_{L_{\ell}}, \qquad \forall \ \ell. \tag{3}$$

Equation (3) determines the demand for labor given the real wage. In the first stage of the decision process, firms and employees set wages such that they decline if the unemployment rate increases. This enables us to write the aggregate wage-setting equation as

$$w_{\ell} = \phi_{\ell} \left(u_{\ell} \right), \qquad \phi_{\ell}' < 0, \qquad \forall \ \ell, \tag{4}$$

where ϕ_{ℓ} is a function that captures the response of the wage to the unemployment rate $u_{\ell} = 1 - L_{\ell}/N_{\ell}$. The rationale behind equation (4) is that a higher unemployment rate weakens the outside options of workers and, hence, reduces their wages. As outlined above, this is consistent with a range of wage-setting models including right to manage collective bargaining models and efficiency wage models.

The wage-setting relation in equation (4) and the relation between the real wage and marginal product of labor in equation (4) allow us to solve for the employment response to a change in foreign labor supply. This requires solving a system of equations which is determined by the wage-setting curves and the production function for each type of labor. This system has to satisfy, in each cell of the labor market, the implicit function

$$\Omega_{\ell}(\mathbf{L}, M) \equiv \nu^{-1} Y_{L_{\ell}}(\mathbf{L}, K(\mathbf{N}(M))) - \phi_{\ell}(u_{\ell}(L_{\ell}, N_{\ell}(M))) = 0, \quad \forall \ell. (5)$$

Note that equation (5) implies that the capital stock may adjust to labor supply shocks, i.e., that $\partial K/\partial \mathbf{N} \geq 0$. Differentiating this system implicitly with respect to a marginal labor supply shock through immigration yields for the change in employment

$$\frac{d\mathbf{L}}{dM} = \left(\frac{\partial\nu^{-1}\mathbf{Y}_{\mathbf{L}}}{\partial\mathbf{L}} - \frac{\partial\mathbf{f}}{\partial\mathbf{u}}\frac{\partial\mathbf{u}}{\partial\mathbf{L}}\right)^{-1} \times \left(\frac{\partial\mathbf{f}}{\partial\mathbf{u}}\frac{\partial\mathbf{u}}{\partial\mathbf{N}}\frac{d\mathbf{N}}{dM} - \frac{\partial\nu^{-1}\mathbf{Y}_{\mathbf{L}}}{\partial K}\frac{\partial K}{\partial\mathbf{N}}\frac{d\mathbf{N}}{dM}\right), \quad (6)$$

where $\mathbf{Y}_{\mathbf{L}}$ denotes a vector of the marginal products of labor in each cell of the labor market as outlined in equation (3), \mathbf{f} a vector of functions that determines the wage response to the unemployment rate as outlined in equation (4), and \mathbf{u} the vector of unemployment rates.

Having solved for the equilibrium employment response, it is straightforward to use the relation in equation (3) for deriving the wage effects of migration:

$$\frac{d\mathbf{w}}{dM} = \frac{\partial \nu^{-1} \mathbf{Y}_{\mathbf{L}}}{\partial \mathbf{L}} \frac{d\mathbf{L}}{dM} + \frac{\partial \nu^{-1} \mathbf{Y}_{\mathbf{L}}}{\partial K} \frac{\partial K}{\partial \mathbf{N}} \frac{d\mathbf{N}}{dM}.$$
(7)

2.2 Outline of the empirical framework

For the empirical analysis, we have to impose more structure on the economy. Similar to Borjas (2003) and Ottaviano/Peri (2006), we follow Card/Lemieux (2001) in using a nested CES production function. More specifically, we employ a four-level production function which groups the workforce into $q = 1, \ldots, 4$ education groups, $j = 1, \ldots, 8$ experience groups, and k = 1, 2 nationality groups.¹ Although the nested CES function imposes some restrictions on the elasticities of substitution, it has the advantage that it is parsimonious in the parameters.

Suppose that aggregate production in equation (1) can be represented by a standard Cobb-Douglas production function:

$$Y_t = A_t L_t^{\alpha} K_t^{1-\alpha}, \tag{8}$$

where Y_t denotes aggregate output, A_t an exogenous parameter which captures total factor productivity, L_t the aggregate labor input, K_t physical capital, α the production elasticity of labor, and t the time index.

Grouping the labor force by education, experience, and national origin yields:

¹Our four-level framework resembles the specification by Ottaviano/Peri (2006). D'Amuri et al. (2008) have recently applied a five-level framework to Germany that distinguishes between different vintages of immigrants in a further nest of the production function. They find that old and new arrivals are perfect substitutes.

$$L_{t} = \left[\sum_{q=1}^{4} \theta_{qt} L_{qt}^{(\delta-1)/\delta}\right]^{\delta/(\delta-1)}, \qquad \sum_{q=1}^{4} \theta_{qt} = 1,$$
(9)

$$L_{qt} = \left[\sum_{j=1}^{8} \theta_{qj} L_{qjt}^{(\rho-1)/\rho}\right]^{\rho/(\rho-1)}, \qquad \sum_{j=1}^{8} \theta_{qj} = 1,$$
(10)

$$L_{qjt} = \left[\sum_{k=1}^{2} \theta_{qjk} L_{qjk}^{(\sigma_q - 1)/\sigma_q}\right]^{\sigma_q/(\sigma_q - 1)}, \ \sum_{k=1}^{2} \theta_{qjk} = 1,$$
(11)

where the aggregate L_t incorporates the contributions of workers who differ in both education and experience, L_{qt} is a labor composite that aggregates all workers with education q, L_{qjt} a labor composite that aggregates native and migrant workers of education q and experience j, and L_{qjkt} the number of employed workers of education q, experience j, and national origin k. The technology parameters θ_{qt} , θ_{qj} , and θ_{qjk} determine the productivity levels of the respective factor. We allow the productivity parameter θ_{qt} to vary over time since skill-biased technological progress might affect the productivity of various types of labor in different ways (Katz/Murphy 1992). The other production parameters are assumed to be constant over time.

Finally, $\delta > 0$, $\rho > 0$, and $\sigma_q > 0$ are constant parameters measuring the elasticity of substitution between labor of different educational levels, between workers with similar education but different work experience, and between native and migrant workers with similar education and experience levels. Our a priori expectation is that workers within each experience group are closer substitutes than those across skill groups, which implies that $\rho > \delta$.

Whether foreign and native workers in each education and experience group are imperfect substitutes is the subject of some controversy in the literature. We therefore test empirically whether native and foreign workers are imperfect substitutes.

Based on equation (3) we can write the real wage rate as the marginal product of labor divided by the mark-up factor. Using the nested CES production function we thus write the log wage of a worker with skill q, experience j and national origin k as

$$\ln w_{qjkt} = \ln \left(\nu^{-1} \alpha A_t^{1/\alpha}\right) + \frac{1}{\delta} \ln L_t + \ln \theta_{qt} - \left(\frac{1}{\delta} - \frac{1}{\rho}\right) \ln L_{qt}$$
(12)
+
$$\ln \theta_{qj} - \left(\frac{1}{\rho - \frac{1}{\sigma_q}}\right) \ln L_{qjt} + \ln \theta_{qjk} - \frac{1}{\sigma_q} \ln L_{qjkt} + \frac{1 - \alpha}{\alpha} \ln \kappa_t ,$$

where $\kappa_t \equiv K_t/Y_t$ denotes the capital-output ratio.

To calculate the wage effects of a labor supply shock due to immigration, we first compute the employment effects. The general solution for the employment effects is given in equation (6), and an explicit solution for our case with 64 types of labor and a nested CES production function is provided in Appendix A1. In the second step, we differentiate the wage equation (12) with respect to the employment changes in all cells of the labor market and with respect to a change in the capital-output ratio triggered by immigration. This gives

$$\frac{dw_{qjkt}}{w_{qjkt}} = \frac{1}{\delta} \sum_{z} \sum_{x} \sum_{m} \left(s_{zxmt} \frac{dL_{zxmt}}{L_{zxmt}} \right)$$

$$- \left(\frac{1}{\delta} - \frac{1}{\rho} \right) \frac{1}{s_{qt}} \sum_{x} \sum_{m} \left(s_{qxmt} \frac{dL_{qxmt}}{L_{qxmt}} \right)$$

$$- \left(\frac{1}{\rho} - \frac{1}{\sigma_q} \right) \frac{1}{s_{qjt}} \sum_{m} \left(s_{qjmt} \frac{dL_{qjmt}}{L_{qjmt}} \right)$$

$$- \frac{1}{\sigma_q} \left(\frac{dL_{qjkt}}{L_{qjkt}} \right) + \frac{1 - \alpha}{\alpha} \frac{d\kappa_t}{\kappa_t} ,$$
(13)

where z = 1, ..., 4 indexes education, x = 1, ..., 8 work experience, and m = 1, 2 national origin, and s denotes the share of wage sum paid to workers in the respective labor market cell in the total wage bill, i.e.,

$$s_{qjkt} = \frac{w_{qjkt}L_{qjkt}}{\sum_{z}\sum_{x}\sum_{m}w_{zxmt}L_{zxmt}},$$

$$s_{qjt} = \frac{\sum_{m}w_{qjmt}L_{qjmt}}{\sum_{z}\sum_{x}\sum_{m}w_{zxmt}L_{zxmt}},$$

$$s_{qt} = \frac{\sum_{x}\sum_{m}w_{qxmt}L_{qxmt}}{\sum_{z}\sum_{x}\sum_{m}w_{zxmt}L_{zxmt}}.$$

3 Data and descriptive evidence

In our empirical analysis we use the IAB employment sample (IABS), a two percent random sample of all German employees registered with the social security system during the period 1975-2004. In addition to socio-economic and job characteristics, the IABS provides information on benefit recipients at the individual level.² The IABS is stratified according to nationality and therefore representative of the native and foreign working population. The data set is especially useful for analyses that take wages into account since the wage information is used to calculate social security contributions and is therefore highly reliable.³

Nevertheless the IABS also has some limitations in the context of our analysis, the main one being that we can identify foreigners only on the basis of citizenship. Some further shortcomings arise from the wage and qualification information provided by the data set.

First, there is no information on the year when immigrants entered the country. Due to the *jus sanguinis* tradition of the German law, naturalization rates are traditionally very low, such that second- and third-generation migrants often still have foreign citizenship and therefore appear as foreign workers in our sample. On August 1, 1999, a new immigration act came into effect that allows German-born children of foreign-born parents who have been living in Germany for at least eight years to decide which nationality to adopt up to the age of 23. This has slightly increased the naturalization of German-born individuals with a migrant background. To mitigate the possible effects of naturalization, we have classified all individuals as foreigners who are reported as foreign citizens in their first available spell. This prevents naturalizations from being displayed as a declining foreigner share in our sample.

Second, ethnic Germans – so-called "Sptaussiedler" – are reported as Germans since the concept of citizenship does not allow us to distinguish between citizens born in Germany and those born abroad. However, language courses and other integration subsidies offered to ethnic Germans should facilitate their labor market integration. These programs are reported in the benefit recipient file added to our data set. This allows us to identify the majority of ethnic Germans who have entered the German labor force since 1980. In our sample, the cumulative inflow of ethnic Germans accounts for

 $^{^{2}}$ About 90 percent of the registered unemployed are eligible for benefits (Wagner/Jahn 2004). Therefore the unemployment rate is only slightly downward-biased.

³In our sample the average size of the foreigner cells is well above 1,000 observations. Therefore it is not likely that our results suffer from an "attenuation bias" (Aydemir/Borjas 2006).

3.2 percent of the labor force in Western Germany.⁴ Since ethnic Germans' labor market performance and language command resembles that of other foreigners (see Glitz 2006), we have classified ethnic Germans as foreigners.

Third, the IABS included Eastern Germany for the first time in 1992. Solely focusing on the unified Germany would exclude the main immigration shock associated with the fall of the iron curtain. German reunification also requires excluding Western Berlin, since mobility between Eastern and Western Berlin has been high since the fall of the Berlin wall in 1989. We therefore concentrate in our analysis to individuals who were employed or unemployed in West Germany on September 30 of any year in the period 1980 to 2004. We do not believe that the focus on Western Germany should significantly affect our results as four-fifth of the German labor force work in Western Germany and the foreigner share is negligible in Eastern Germany.

Fourth, the data set reports gross daily wages and does not provide information on hours worked. We therefore exclude part-time employees, trainees, interns, and at-home workers from the sample since the wage information is not comparable for these groups. For the same reason we exclude workers with wages below the social security contribution threshold.

Fifth, there is some empirical evidence of differences in the early retirement behavior between German and immigrant men (Bonin et al. 2000). We therefore restrict our analysis to individuals between the ages of 15 and 60.

Sixth, our data are right-censored since gross wages can only be observed up to the social security contribution ceiling. About four percent of the employment spells in the final data set are right-censored. This may affect the estimation of the wage-setting curves, particularly in the high-skilled segments of the labor market. We have therefore imputed wages above the social security contribution ceiling using a heteroscedastic single imputation approach specifically developed for the IABS data set (Büttner/Rässler 2008).⁵

Seventh, self-employed workers and civil servants do not contribute to the social security system and are therefore not covered by our sample. While the self-employment rate of natives increased only slightly from 9.5 percent in 1985 up to 10 percent in 2000, the self-employment rate of immigrants increased from about 6 percent to about 9 percent (Kontos 2003). Nevertheless, the change has been moderate and we do not expect that this will bias our results considerably. In the case of civil servants, it seems plausible to assume that due to legal restrictions, immigrants do not displace natives.

⁴Appendix B1, which is available upon request, provides information on the stock and flow of ethnic German workers into the German labor market.

⁵A more detailed description of the imputation procedure is provided in Appendix B2, which is available upon request.

Eighth, the information on education is provided by employers. This means that information on educational levels is missing for about 17 percent of the individuals. Foreigners are disproportionately affected by missing information on educational levels. We therefore imputed the missing information on education by employing a procedure developed by Fitzenberger et al. (2005), which allows inconsistent education information to be corrected over time as well. After applying this imputation procedure, we had to drop only 1.6 percent of the individuals due to missing or inconsistent information on education (see Appendix A2).

Finally, education and work experience acquired in foreign countries may not have the same value in the labor market as education and experience obtained in Germany. Moreover, certain characteristics of foreigners, such as their command of the German language, may prevent them from fully transferring their human capital to the German labor market. However, the classification of educational levels in the IABS reflects the official recognition of educational degrees acquired abroad, which is rather restrictive in Germany. As a consequence, the correlation between educational degree and occupational status is similar for foreigners and natives in our data set (in fact, slightly higher for foreigners). In the highest education group, i.e., individuals with a university degree, the share of immigrants in high-level occupations is only slightly below that of natives (see Glitz 2006 for a similar result). Note also that in their analysis of wage elasticities D'Amuri et al. (2008) find no differences, whether classifying the labor force by educational or by occupational level.

As the above sample selection mainly affects groups in the labor market where foreigners are under-represented (civil servants, the self-employed, part-time workers), we might slightly overstate the impact of legal immigration in our analysis. However, one caveat might be that we are not able to observe illegal immigrant workers, who may exert pressures on the less-skilled segments of the labor market in particular.

Following the model outlined in Section 2, we group the labor force by education and potential work experience. A sensible classification following the characteristics of the German labor market requires us to distinguish four educational groups: no vocational degree, vocational degree, high school degree ("Abitur") with vocational degree, and university degree. At first glance, one might consider aggregating the groups "vocational degree" and "high school degree with vocational degree", but in Germany these are separate labor markets. Despite the small size of the group with a high school degree, we therefore decided to treat it separately.

Furthermore, we distinguish eight potential work experience classes following the standard approach by Borjas (2003), subtracting the typical number of years spent in the educational system from the age of the worker and splitting experience into five-year intervals.

Figure 1 displays the share of foreigners - including ethnic Germans - in the labor force and the share of foreigners in the employed workforce. During the 1980s we observe a sharp decline, which is a consequence of tightening migration restrictions in the wake of the first and second oil price shock. Between the mid 1980s and the mid 1990s, the foreign share in the labor force increased by 5 percentage points. The sharp increase in the foreigner share during the 1990s resulted from the collapse of the Central and Eastern European communist regimes and the civil wars in the former Yugoslavia, which triggered large-scale migration to Germany. Since the early 2000s, the foreigner share has plateaued due to the slowdown in economic growth and tighter restrictions on immigration. Moreover, since foreigners tend to be more than proportionally affected by unemployment, their share in the employed workforce declined relative to their share in the labor force at the end of the millennium.

Figure 1 about here

Table 1 shows that the increase in the foreign labor supply disproportionally affects the higher education groups: while the number of foreign workers without a vocational degree declined by one-fourth during the 1980-2004 period, the number of foreigners with a university degree increased by 164 percent, and the number of foreigners with a high-school and vocational training degree by a factor of 6. The largest education group — those with a vocational degree — increased by 134 percent, while the total foreign labor force increased by 40 percent.

Table 1 about here

4 Estimation

4.1 Wage-setting equations

The first step in the empirical application of the model outlined in Section 2 is to estimate the wage-setting equations. Following Bell et al. (2002) and

Blanchflower/Oswald (2005) we estimate the elasticity of the wage-setting curve in dynamic form, i.e., as

$$\ln w_{qjt} = \beta_{qj} \ln w_{qj,t-1} + \gamma_{qj} \ln u_{qjt} + \lambda_{qj} \tau_{qjt} + \eta' \mathbf{X}_t + e_{qjt} , \qquad (14)$$

where γ_{qj} denotes the short-run elasticity between the wage and the unemployment rate, \mathbf{X}_t a vector of control variables, η the corresponding vector of coefficients, and τ_{qjt} an education-experience-specific deterministic time trend. We include a linear and a squared trend here. As controls for macroeconomic shocks we use real GDP, the consumer price index, and the industrial production index.⁶ The error term e_{qjt} is specified as a one-way error component model with fixed effects for each education-experience group.

Unobserved shocks may affect wages and the unemployment rate simultaneously. In order to address this problem, we estimate equation (14) by 2SLS. We use three instruments for the unemployment rate. First, following Bartik (1991) and Blanchard/Katz (1992) we employ an industry mix variable that measures how much of the deviation in employment growth in an education-experience cell from average employment growth can be explained by a concentration of workers in the respective cell in fast- or slow-growing industries. This variable simply measures how much of the change in employment can be attributed to an exogenous shift of the sectoral structure. Our second instrument is an export demand index, which is constructed as the log of the GDP per capita at constant prices and exchange rates of all OECD countries weighted by their average share in German exports during the sample period. This variable should capture exogenous shifts in labor demand that are triggered by the economic activity of Germany's trading partners.⁷

Third, we instrument the unemployment rate with a potential immigration variable, which measures potential exogenous labor supply shocks due to immigration in each education-experience cell of the labor market. Since immigration itself might be endogenous, we estimate auxiliary regressions that explain the stock of foreign workers in each education-experience cell by push factors in the sending countries (log GDP per capita, log unemployment rate, institutional variables that capture political shocks and migration conditions) and bilateral fixed effects. The coefficients of these regressions are used to calculate the migration potential in each education-experience group. By construction, this variable captures labor supply shocks due to immigration that are driven by exogenous factors. Several diagnostic tests

⁶See Appendix A3 for a description of the variables.

 $^{^{7}}$ A similar instrument has been used by Carlsson et al. (2008) in the estimation of an employment equation.

support the hypothesis that our instruments are valid and relevant (see notes in Table 2).

The specification of equation (14) is similar to that used in the wagesetting and wage curve literature, but it differs from the usual approach in that it allows the elasticity between wages and the unemployment rate to differ across education-experience groups. However, we aggregate native and foreign workers.⁸ This reflects not only the difficulties in empirically identifying the elasticities of the wage-setting curve for a rather small group like foreigners, but also the fact that collective wage agreements determine a wage floor for education and experience groups but do not discriminate a priori between native and foreign workers. Moreover, the German antidiscrimination legislation reduces opportunities to set wages differently for native and foreign workers.⁹

For the identification of the coefficients in equation (14) we exploit the long time dimension of our data set. The underlying assumption that the coefficients are stable over time may be questionable if the wage-setting mechanism changes. In the case of Western Germany, the number of employees covered by industry- or firm-level collective agreements did indeed decline by 11 percentage points between 1996 and 2004. However, most firms that left the employer federations still apply wage levels fixed in collective agreements, such that the share of employees who are directly or indirectly covered by collective wage agreements remained rather stable at about 85 percent of the workforce in Western Germany (Ellguth/Kohaut 2007).¹⁰

Table 2 about here

The results are displayed in Table 2. All regressions have the expected negative sign for the coefficient on the unemployment rate. All short-term elasticities are significant and with one exception, the long-term elasticities are also significant. The autoregressive parameter on the lagged wage is well below 1, supporting a wage-setting curve rather than a Phillips curve.¹¹

⁸In each regression we have pooled two experience groups to achieve more stable results. ⁹We can hardly test for this assumption since we estimate point elasticities and since the unemployment rate differs between natives and foreigners.

¹⁰We conducted Wald tests to test for potential structural breaks that might be caused by changes in the wage-setting mechanism at the middle of the sample period. The test statistics cannot reject the null hypothesis of no structural breaks.

¹¹In one case we obtain a negative but insignificant coefficient for the lagged wage. In our simulations we assume that adjustment takes place immediately, in this case by setting this coefficient to zero.

The first regressions provide pooled estimates of the wage-setting curve for all groups and for each education group separately. In the regression where all education-experience groups are pooled, we find a short-run elasticity of about -0.08 and a long-run elasticity of about -0.16. The national-level estimates presented here are somewhat higher than the average elasticity of -0.1 found by the regional-level wage curve literature in other OECD countries (Blanchflower/Oswald 1994, Nijkamp/Poot 2005), but much higher than the elasticity of -0.03 estimated by Baltagi et al. (2009) at the regional level in Germany. This is not surprising since the regional level estimates control for all macroeconomic influences that are particularly relevant in economies such as Germany, where industry-level bargaining plays an important role.

The most intriguing finding in our disaggregated estimates is that of very high elasticities of the wage-setting curve in the segments with little work experience, with the notable exception of the group without vocational training. This indicates that seniority wage considerations play an important role in wage-setting in the skilled segments of the labor market. Note that this finding is relevant in our context since most newly arrived immigrants possess little work experience. Interestingly enough, we do not obtain a lower elasticity of the wage-setting curve in the group where union density and the coverage rate of collective wage agreements is particularly high in Germany: individuals with vocational training. In contrast, the aggregate elasticities of the wage-setting curve are, at values of between -0.15 and -0.19, relatively similar across educational groups.

4.2 Labor demand equations

In the next step we estimate the elasticities of substitution between the different types of labor based on our nested production function framework. Our identification strategy differs from the one used in the literature, which relies on perfect competition with clearing labor markets (Card/Lemieux 2001, Borjas 2003, Ottaviano/Peri 2006, 2008). While this literature treats employment as exogenous and wages as the endogenous variable, it follows from our wage-setting framework that labor demand is endogenously determined once wages are fixed.

Let us start with the identification of the elasticity of substitution between native and foreign workers. Based on equation (12) we can express the relative demand for native and foreign workers with education q and experience j as $\ln (L_{qjht}/L_{qjft}) = \sigma_q \ln (\theta_{qjh}/\theta_{qjf}) - \sigma_q \ln (w_{qjht}/w_{qjft})$, where the index h denotes natives and the index f foreigners. We estimate this equation as

$$\ln\left(L_{qjht}/L_{qjft}\right) = D_{qj} - \sigma_q \ln\left(w_{qjht}/w_{qjft}\right) + \mu' \mathbf{X}_t + \varepsilon_{qjt}, \tag{15}$$

where D_{qj} denotes a vector of dummy variable for each education-experience cell, \mathbf{X}_t a vector of control variables, μ the corresponding vector of coefficients and ε_{qjt} a zero-mean disturbance term. Following the approach of Ottaviano/Peri (2008) the dummy variables in each education-experience cell capture the log of the relative labor productivity of natives and foreigners times the elasticity of substitution. This implies that the relative productivity of natives and immigrants varies across education and experience groups but is constant over time.¹²

As macroeconomic control variables we use real GDP growth, an index of the domestic crude oil price, and the export performance index of the OECD. We do not consider time fixed effects since the model is identified by the variance over time. Note that technology shifts in the productivity of education (or experience) groups and macroeconomic shocks common to both natives and foreigners are absorbed by higher levels of the production function.

Estimating equation (15) by OLS can generate inconsistent results if unobserved idiosyncratic shocks affect both the relative labor demands and relative wages of natives and foreigners. To address this problem we estimate equation (15) with 2SLS. As instruments we first use the ratio of the log of the average level of unemployment benefits for natives and foreigners in each education-experience cell as an instrument for the relative wage levels. Accurate information on the level of unemployment benefits is provided by the IABS. Unemployment benefits are a suitable instrument if they affect the wage level via the reservation wage without moving labor demand directly. Since it may take time for unemployment benefits to affect wages, we use the first and the second lag of the unemployment benefits as instruments.

Our second instrument is an ideology index, which captures the share of left- and right-wing parties in the government weighted by their seats in parliament (Bjrnskov 2008)¹³. This instrument captures governmental policies that can affect reservation wages through different channels, e.g., progressive taxation, generosity of welfare benefits, etc. Note that foreigners' access to welfare benefits is a core issue in the policy debate, which in turn

¹²Borjas et al. (2008) suggest also including interaction dummies of the educationexperience-specific fixed effects with linear time trends, which would absorb a large part of the identifying variation. However, there is no empirical evidence for Germany that the relative labor productivity of foreigners has changed systematically over time at a given level of education and work experience.

¹³We are grateful to Christian Bjrnskov who provided the ideology index.

affects relative reservation wages for foreigners and natives.¹⁴

Table 3 about here

The results of our estimates are reported in Table 3. All coefficients for σ_q are significantly different from zero, providing support for the hypothesis that native and foreign workers are imperfect substitutes. At 7, the overall elasticity between native and foreign workers is similar to that found by Ottaviano/Peri (2006) for the US, but smaller than the elasticity of between 16 and 21 identified by D'Amuri et al. (2008) for Germany based on an identification strategy that assumes clearing labor markets. The elasticity of substitution is particularly high for workers with vocational training and a high school degree, but relatively low both for less skilled workers and workers with a university degree.

In the next step we estimate the elasticity of substitution between experience groups. Using equation (12) once again we can estimate the demand for the labor composite L_{qjt} as

$$\ln \hat{L}_{qjt} = D_t + D_{qt} + D_{qj} - \rho \ln w_{qjt} + v_{qjt} , \qquad (16)$$

where the time-specific fixed effects D_t control for the variance of $\rho \ln \left(\nu^{-1} \alpha A_t^{1/\alpha} \kappa_t^{(1-\alpha)/\alpha}\right) + \frac{\rho}{\delta} \ln L_t$, the time by education-specific fixed effects D_{qt} for the variation in $\rho \ln \theta_{qt} - \left(\frac{\rho}{\delta} - 1\right) \ln L_{qt}$, and the education-experience group fixed effects D_{qj} for the variance in $\rho \ln \theta_{qj}$, i.e., in the productivity term times the elasticity of substitution, which is assumed to be constant over time. v_{qjt} denotes the zero-mean disturbance.

The labor composite is calculated as

$$\hat{L}_{qjt} = \left[\hat{\theta}_{qjh}L_{qjht}^{(\widehat{\sigma}_q-1)/\widehat{\sigma}_q} + \hat{\theta}_{qjf}L_{qjft}^{(\widehat{\sigma}_q-1)/\widehat{\sigma}_q}\right]^{\widehat{\sigma}_q/(\widehat{\sigma}_q-1)},$$

where we use our estimates of the education-experience-specific fixed effects from equation (15) to calculate the productivity parameters for native and foreign workers as $\hat{\theta}_{qjh} = \frac{\exp(\hat{D}_{qj}/\hat{\sigma}_q)}{1+\exp(\hat{D}_{qj}/\hat{\sigma}_q)}$ and $\hat{\theta}_{qjf} = \frac{1}{1+\exp(\hat{D}_{qj}/\hat{\sigma}_q)}$.

¹⁴The regression diagnostics indicate that these instruments are both valid and relevant (see notes to Table 3). In addition, we conducted Wald tests to test for potential structural breaks at the middle of the sample period. The test statistics cannot reject the null hypothesis of no structural breaks.

We estimate equation (16) by 2SLS using the first and second lags of the log of the average unemployment benefit in each education-experience cell and the ideology index as instruments. Our regression diagnostics again suggest that these instruments are valid and relevant. We find an elasticity of substitution across experience groups of 8.6, which is close to other findings in the international literature (Card/Lemieux 2001, Borjas 2003, Ottaviano/Peri 2006).

The elasticity of substitution between education groups is estimated analogously as

$$\ln \hat{L}_{qt} = D_t + D_q + \beta_q \tau_{qt} - \delta \ln w_{qt} + \xi_{qt} \quad , \tag{17}$$

where the time-specific fixed effects D_t control for the variance of $\delta \ln \left(\nu^{-1} \alpha A_t^{1/\alpha} \kappa_t^{(1-\alpha)/\alpha}\right) + \ln L_t$ and other macroeconomic fluctuations, the education-specific fixed effects D_q , and the education-specific deterministic time trend τ_{qt} for the variance in the term $\delta \ln \theta_{qt}$, which captures the variance in the skill-specific productivity parameter which is driven inter alia by skill-biased technological progress (see Katz/Murphy 1992, for a similar specification). ξ_{qt} denotes a zero-mean random disturbance.

The labor composite L_{qt} is computed as

$$\hat{L}_{qt} = \left[\sum_{j=1}^{8} \hat{\theta}_{qj} \hat{L}_{qjt}^{(\hat{\rho}-1)/\hat{\rho}}\right]^{\hat{\rho}/(\hat{\rho}-1)},$$

where the estimated efficiency parameters $\hat{\theta}_{qj}$ are derived from the fixedeffects estimates as $\hat{\theta}_{qj} = \frac{\exp(\hat{D}_{qj}/\hat{\rho})}{\sum_{j} \exp(\hat{D}_{qj}/\hat{\rho})}$.

We estimate equation (17) by 2SLS using the log of first and second lags of the average unemployment benefits in each education group as an instrument, which again turns out to be valid and relevant. We find an elasticity of 3, which is similar to that found in the US literature (Katz/Murphy 1992, Ottaviano/Peri 2006), but below what other studies find for Germany (Felbermayr et al. 2008).

5 Simulation of the immigration impact

We now use the estimated parameter values to simulate the impact of migration on (un)employment and wages. We simulate two scenarios here.

First, we simulate the effects of a one percent increase in the labor force due to immigration using the education and experience composition of the foreign workforce at the average of the sample period. This scenario provides an indication as to the marginal effects of immigration at the given structure of the foreign workforce.

Second, we simulate the annual averages of the wage and employment effects of the actual immigration shock for the entire 1980-2004 period. During this period the foreign labor force increased by 40 percent, which corresponds to an increase of four percent of the total labor force. Since this is an inframarginal labor supply shock in many cells of the labor market, we calculate annual averages for this period. For this purpose we first simulate the annual wage and unemployment effects based on the actual changes in each cell of the labor market for each year and then calculate the annual average for the 1980-2004 period using the share of the wage sum and of the labor force in each education and experience cell in each year as weights.¹⁵

In the one percent scenario, we distinguish between the short-run and the long-run effects of immigration. In the simulations of the short-run impact we use the short-term elasticities of the wage-setting curves as estimated in equation (14) and assume that the capital stock remains fixed. In the long-run simulations we use the long-term elasticities of the wage-setting curves and assume that the capital stock adjusts completely to an aggregate labor supply shock, such that the capital-output ratio is fixed. The latter assumption is empirically supported by the Kaldor facts on economic growth and can be explained, inter alia, by international capital mobility (see Ottaviano/Peri 2006, for a discussion). Finally, as a robustness check, we have calculated the effects of immigration assuming clearing labor markets, i.e. an economy where the elasticity of the wage-setting curve approaches infinity and unemployment is zero.

In the scenario that covers the 1980-2004 period we present the long-run scenario only, since a short-term scenario does not make sense for such a long time period.

In all scenarios, we start with the calculation of the change in employment based on equation (6). The explicit form of the vectors and matrices that we employ on the basis of our nested production function is provided in Appendix A1. The parameters for the wage-setting curves are taken from our estimates of equation (14) and the parameters σ_q , ρ , and δ from our estimates of equations (15) – (17). Following the literature, we set α to 0.67 (Cahuc/Zylberberg 2001). Having calculated the employment effects of immigration, we use equation (12) for the calculation of the wage effect. The

¹⁵Appendix B3, which is available upon request, also provides the results dividing the observation period in five year intervals. The results do not differ fundamentally, except for the subperiods 1985-1989 and 1990-1994, where the immigration shock has been particularly large. For these two intervals we find that the unemployment rate increases and wages drop in all foreign education groups.

shares of education and education-experience groups in the total wage bill are taken from our data set. Note that the mark-up factor is a constant that cancels out when we calculate changes of wage and (un)employment levels.

Table 4 reports the average effects for the total labor force, the native labor force, and the foreign labor force by educational levels. For the calculation of the average effects, we weight the wage changes by the income share in each cell, and the changes in the unemployment rate by the share in the labor force in each cell.

Effects of a one percent increase: Our simulation results indicate that an immigration of one percent of the labor force at the average skill and experience structure of the foreign workforce during the sample period reduces overall average wages by 0.18 percent and increases the average unemployment rate by 0.31 percentage points in the short run. The pre-existing foreign labor force bears the brunt of adjustment: their wages decline by 1.11 percent and their unemployment rate increases substantially by almost two percentage points in the short run. In contrast, the native workforce is only slightly affected. Their wages decline by 0.08 percent and their unemployment increases by 0.09 percent in the short run.

The large difference in the labor market effects for the foreign and the native workforce can be traced back to two main facts: First, the elasticity of substitution between native and foreign workers is relatively low, which implies that the labor supply shocks in specific education and experience cells of the labor market can spill over to native workers only to a limited extent. The particularly low elasticity of substitution between foreigners and natives in the education groups without vocational training and with a university degree is also responsible for the fact that the foreign labor supply shock mainly affects the foreign workforce in these two groups.

Second, the skill and experience structure of the immigrant influx resembles that of the foreign workforce, but differs substantially from that of the native workforce in this scenario. Since the education and experience groups are imperfect substitutes, this again implies that the foreign workforce is affected more than the native workforce. At the aggregate level, we find that the education group without vocational training suffers particularly from the labor supply shock since immigrants are represented disproportionally in this skill group.

The picture looks brighter if we consider the long-run effects of immigration. The adjustment of capital stocks ameliorates the labor supply shock, such that the average wage level of the total labor force remains constant. The unemployment rate increases by 0.08 percentage points in the long-run simulations, which can be explained by the fact that in this scenario, immigrants enter labor market cells with high unemployment rates and rather low elasticities of the wage-setting curves.

As a consequence of the low elasticity of substitution between natives and foreigners and the skill composition of the labor supply shock, the native labor force benefits in terms of higher wages (0.11 percent) and lower unemployment (-0.06 percentage points) in the long-run scenario, while foreign workers suffer from lower wages (-1.09 percent) and higher unemployment (1.16 percentage points). Again, the pre-existing foreign workforce is particularly affected in those education segments of the labor market where the elasticity of substitution between natives and foreigners is low.

Comparing these results with the counterfactual case of an economy with clearing labor markets shows that wage rigidities play an important role in protecting employed workers against wage competition from immigrants: The wage effects of immigration would increase by a factor of about 1.55 in the counterfactual case of clearing labor markets compared to the case with wage rigidities in the short-term scenario. The role of wage rigidities is mitigated, however, in the long-term scenario when capital stocks adjust: The wage effects in the clearing labor market scenario exceed those in the scenario with wage rigidities by a factor of 1.3 in the long-run according to our simulations.

Table 4 about here

Effects of the 1980-2004 immigration shock: In contrast to the previous scenario, the actual immigration during the 1980-2004 period is associated with a decline in the group of no vocational training, while the number of immigrants in the other skill groups increased continuously. This yields a different picture.

The most intriguing finding in this scenario is that immigration did not affect the unemployment rate in Germany. While the unemployment rate of natives and the total labor force remained constant over the period, the unemployment rate of the pre-existing foreign labor force increased only marginally – by less than 0.01 percentage points per annum. While the increasing labor supply of foreigners raised unemployment rates in the cells that were affected most by the influx, the cross-effects of the labor supply shifted the increased labor demand to other cells, such that the overall effects cancel one another out in this scenario. This is because foreign labor supply declined in the labor market segment with the highest unemployment rate and a rather low elasticity of the wage setting curve – i.e., individuals without vocational training – while increasing in the higher-skilled segments of the labor market, where unemployment is lower and the elasticity of the wage-setting curve is higher.

The wage levels of the native labor force increased slightly $(+0.01 \text{ per$ $cent p.a.})$ while those of the foreign workforce declined slightly (-0.09 percentp.a.). We observe substantial increases in the wages of foreign workers without vocational training, while those of foreign workers with a university and a high school degree declined substantially as a consequence of the high labor influx in these skill groups. The native labor force benefits both at the lower and higher end of the skill spectrum. At the level of the entire labor force, we find that wage levels increased in the group without vocational training, and declined slightly in all other education groups, which corresponds again to the skill composition of the labor supply shift.

Comparing our findings: The elasticity between wages and immigration in our one percent scenario is about half the size of that found by Borjas (2003) and Ottaviano/Peri (2006, 2008) for the US and slightly below what Manacorda et al. (2006) and Dustmann et al. (2008) find for the UK. This is hardly surprising since these studies derive the wage effects of immigration assuming clearing labor markets, whereas we consider wage rigidities.

Comparing our findings with those of D'Amuri et al. (2008) provides some interesting insights into the differences between their outcomes and ours resulting from the different methodological approaches used to address the unemployment effects of immigration. Their reduced-form estimates of the employment impact of immigration suggest that a ten percent increase in the foreign workforce increases the unemployment rate of foreign workers by between 1.5 and two percentage points, while the native workforce remains unaffected.

This is consistent with our finding that a one percent increase of the labor force through immigration – which corresponds roughly to a ten percent increase in the foreign labor force – increases the unemployment rate of foreigners by two percentage points in the short term at the given skill structure of the foreign workforce.

However, while the partial correlation approach of D'Amuri et al. (2008) suggests that the actual migrant influx into Germany increased the unemployment rate of the foreign workforce substantially, we find that the unemployment rate of the foreign workforce remained (almost) constant over the entire sample period. This can be traced back to the fact that the increasing labor supply in the higher skilled segments of the labor market reduced unemployment in the less-skilled segments, which are characterized by lower wage flexibility. The decline in unemployment in these cells exceeded the replacement effects in the high-skilled cells of the foreign labor force. The consideration of these cross-effects in our approach thus delivers a differ-

ent picture of the unemployment effects of the actual immigrant influx into Germany.

6 Conclusions

Concerns about immigration affecting not only wages but also native employment opportunities are widespread in continental Europe, where labor market rigidities are prevalent. In this paper we present a general equilibrium framework that allows us to analyze the wage and employment effects of migration simultaneously in a setting with imperfect labor markets. Our empirical findings suggest that the wage flexibility varies widely for different segments of the labor market. The elasticity of the wage-setting curve is particularly high for workers with little work experience, i.e. labor market segments where newly arrived immigrants are more than proportionally represented.

Our approach provides a number of new insights. In contrast to the literature studying the employment effects of immigration based on partial correlations between the (un)employment and the immigrant rate in certain segments of the labor market, we find that immigration can either raise or reduce unemployment depending on the education and experience structure of the immigrant influx and the wage flexibility in different segments of the labor market. According to our simulations, the immigration of about four percent of the labor force during the 1980-2004 period did not increase either the aggregate unemployment rate of the workforce or the unemployment rate of the foreign labor force. This can be traced back to the fact that the average skill level of the immigrant workforce has increased substantially over time. The higher labor supply in labor market segments with higher wage flexibility and lower unemployment has created an additional demand for less-skilled workers, which compensates for the replacement effects in other education and experience cells of the labor market.

Another intriguing finding is the strong evidence that native and foreign workers are imperfect substitutes in the labor market. As a consequence, the native workforce tends to benefit from immigration in terms of higher wages and lower unemployment risks in all simulations, at least in the long run, although these effects are small. In contrast, the immigration of foreign workers has a major impact on the foreign labor force. While the foreign workforce would suffer substantially from immigration at the given skill and experience structure, our results also show that the foreign workers can benefit if new immigrants are high-skilled, since immigrants are generally more than proportionally represented in the less-skilled segments of the labor market.

Our findings have some important policy implications. Selection of immigrants by human capital characteristics such as education and age is a crucial issue in economies that suffer from wage and other labor market rigidities. In the case of Germany, the gains from immigration are particularly large if immigrants are educated and if they are young, since the flexibility of the labor market is high in these segments. Moreover, policy measures that attempt to increase the elasticity of substitution between native and foreign workers, e.g., through improved labor market integration and efforts to facilitate the transfer of human capital, would mitigate the polarization of wages and employment opportunities of native and foreign workers. Although such policies would reduce the gains of the native labor force from immigration in terms of higher wages and lower unemployment, they would increase social cohesion and reduce the potential costs of immigration that arise through the welfare state channel in the receiving countries.

References

Aydemir, A. and Borjas, G. J. (2006), Attenuation Bias in Measuring the Wage Impact of Immigration," Mimeo, Harvard University.

...... (2007), Cross-Country Variation in the Impact of International Migration: Canada, Mexico and the United States, *Journal of the European Economic Association* 5, 663–708.

Baltagi, B. H.; Blien, U. and Wolf, K. (2009), New evidence on the dynamic wage curve for Western Germany: 1980–2004, *Labour Economics* 16, 47-51.

Bartik, T. (1991), Who Benefits from State and Local Economic Development Policies. Kalamazoo, MI, Upjohn Institute.

Bell, B.; Nickell, S. and Quintini, G. (2002), Wage equations, wage curves and all that, *Labour Economics* 9, 341–360.

Bentolila, S.; Dolado, J. J. and Jimeno, J. F. (2008), Does Immigration Affect the Philipps Curve? Some Evidence for Spain, European Economic Review 52, 1398–1423.

Binyamini, A. and Razin, A. (2008), Inflation-Output Tradeoff as

Equilibrium Outcome of Globalization, Israel Economic Review 6, 109-134

Bjørnskov, C. (2008), The growth-inequality association: Government ideology matters *Journal of Development Economics* 87, 300–308.

Blanchard O. J. and Katz, L. (1992): Regional Evolutions, *Brookings* Papers on Economic Activity 1, 1-75.

Blanchflower, D.D. and Oswald, A. J. (1994), *The wage curve*. Cambridge MA.

(2005): The wage curve reloaded, NBER Working Paper 11338.

Boeri, T. and Brücker, H. (2005), Why are Europeans so tough on migrants?, *Economic Policy* 44, 631–703.

Bonin, H.; Raffelhüschen, B. and Walliser, J. (2000), Can Immigration Alleviate the Demographic Burden, *FinanzArchiv* 57, 1–21.

Borjas, G. J. (2003), The Labor Demand Curve is Downward Sloping: Reexamining the Impact of Immigration on the Labor Market, *Quarterly Journal of Economics* CXVIII, 1335–1374.

Borjas, G. J.; Grogger, J. and Hanson, G. H. (2008), Imperfect Substitution between Immigrants and Natives: A Reappraisal, NBER Working Paper 13887.

Büttner, T. and Rässler, S. (2008), Multiple imputation of right-censored wages in the German IAB employment register considering heteroscedasticity, in *Federal Committee on Statistical Methodology Research Conference* 2007, ed. by F. C. on Statistical Methodology of the United States. Arlington.

Cahuc, P. and Zylberberg, A. (2001), The consequences of the minimum wage when other wages are bargained over, *European Economic Review* 45, 337-352.

Card, D. (1990), The impact of the Mariel boatlift on the Miami labor market, *Industrial and Labor Relations Review* 43, 245-57.

_____ (1995), The wage curve: a review, Journal of Economic Literature

33, 785-799.

Card, D. and Lemieux, T. (2001), Can Falling Supply Explain the Rising Return to College for Younger Men? A Cohort Based Analysis, *Quarterly Journal of Economics* CXVI, 705–746.

Carlsson, M.; Eriksson, S. and Gottfries, N. (2008), How Important are Search Frictions for Employment Dynamics?, mimeo, 23 October 2008.

D'Amuri, F.; Ottaviano, G., and Peri, G. (2008), The Labor Market Impact of Immigration in Western Germany in the 1990's, NBER Working Paper 13851.

Dustmann, C.; Fabbri, F. and Preston, I. (2005), The Impact of Immigration on the British Labor Market, *Economic Journal*, 115, F324–F341.

Dustmann, C.; Frattini, T. and Preston, I. (2008), The Effect of Immigration on the Distribution of Wages, CreAM Discussion Paper 03/08.

Ellguth, P. and Kohaut, S. (2007), Tarifbindung und betriebliche Interessenvertretung - Aktuelle Ergebnisse aus den Betriebspanel 2006, (Union Coverage and Work Councils: Current Evidence from the Establishment Panel 2006), WSI-Mitteilungen 9/2007, 511-514.

Engler, P. (2007), Gains from Migration in a New-Keynesian Framework, mimeo, Freie Universitt Berlin.

Felbermayr, G.; Geis, W. and Kohler, W. (2008), Restrictive Immigration Policies in Germany: Pains and Gains Foregone?, CESifoWorking Paper 2316.

Fitzenberger, B.; Osikominu, A. and Völter, R. (2005), Imputation Rules to Improve the Education Variable in the IAB Employment Subsample, ZEW Discussion Paper 05-10.

Freeman, R. (2006), People Flows in Globalization, *Journal of Economic Perspectives* 9, 22–44.

Glitz, A. (2006), The Labor Market Impact of Immigration: Quasi-Experimental Evidence, CreAM Discussion Paper 12/06. Grossmann, J.B. (1992), The Substitutability of Natives and Immigrants in Production, *Review of Economics and Statistics* 54, 596-603.

Hunt, J. (1992), The impact of the 1962 repatriates from Algeria on the French labor market, *Industrial and Labor Relations Review* 45, 556-72.

Jung, S. and Schnabel, C. (2009), Paying more than necessary? The Wage Cushion in Germany, IZA Discussion Paper 4278.

Katz, L. and Murphy, K. (1992), Change in Relative Wages 1963-1987: Supply and Demand Factors, *Quarterly Journal of Economics* 107, 35–78.

Kontos, M. (2003), Self-employment policies and migrants' entrepreneurship in Germany. *Entrepreneurship and Regional Development* 15, 119-135.

Layard, R. and Nickell, S. J. (1986), Unemployment in Britain, *Economica* 53, Supplement 86, S121–S169.

Layard, R.; Nickell, S. J. and Jackman, R. (2005), *Unemployment: Macroeconomic Performance and the Labour Market, 2nd edition*. Oxford University Press, Oxford.

Levine, P. (1999), The welfare economics of migration control, *Journal* of *Population Economics* 12, 23–43.

Longhi, S.; Nijkamp, P. and Poot, J. (2006), The Impact of Immigration on the Employment of Natives in Regional Labour Markets: A Meta-Analysis, IZA Discussion Paper 2044.

(2008): Meta-Analysis of Empirical Evidence on the Labor Market Impacts of Immigration, IZA Discussion Paper 3418.

Manacorda, M.; Manning, A. and Wadsworth, J. (2006), The Impact of Immigration on the Structure of Male Wages: Theory and Evidence from Britain, IZA Discussion Paper 2352.

Nickell, S. and Andrews, M. (1983), Unions, Real Wages and Employment in Britain 1951-79, Oxford Economic Papers 35, 183-205.

Nijkamp, P. and Poot, J. (2005), The last word on the wage curve? A meta-analytic assessment, *Journal of Economic Surveys* 19, 421–450.

OECD (1994), The OECD Jobs Study, Evidence and Explanations, Vols. I and II, OECD, Paris.

OECD (2009), OECD ANA database, Export performance of goods and services, http://stats.oecd.org.

Ottaviano, G., and Peri, G. (2006), Rethinking the Effects of Immigration on Wages, NBER Working Paper 12497.

(2008), Immigration and National Wages: Clarifying the Theory and the Empirics, NBER Working Paper 14188.

Pischke, J. S. and Velling, J. (1997), Employment effects of immigration to Germany: an analysis based on local labor markets, *Review of Economics* and Statistics 79, 594–604.

Sachverständigenrat (2009), National Time Series, http://www.sachverstaendigenrat-wirtschaft.de.

Salop, S. (1979), A Model of the Natural Rate of Unemployment, American Economic Review 74, 117-125.

Shapiro, C., and Stiglitz, J. E. (1984), Equilibrium unemployment as a worker discipline device, *American Economic Review* 74, 433–444.

Stock, J. H. and Yogo, M. (2005), Testing for weak instruments in linear IV regression, in D.W.K. Andrews and J.H. Stock (eds.), *Identification and Inference for Econometric Models: Essays in Honor of Thomas Rothenberg*, Cambridge University Press, Cambridge.

Wagner, T. and Jahn, E. J. (2004), *Neue Arbeitsmarkttheorien, (New Labor Market Theories)*, 2nd edition. Lucius and Lucius, Stuttgart.

World Bank (2009), World Development Indicators. World Bank, Washington, DC.

A The explicit solution for the employment response

The general solution for the marginal employment response to an increase in labor supply through immigration is given in equation (6). The model in section 2.2 distinguishes $4 \times 8 \times 2 = 64$ types of labor. Using the notation from the nested production function we write the 1×64 vectors as $\mathbf{x} =$ $[x_{111}, x_{112}, x_{121}, \ldots, x_{211}, \ldots, x_{ijk}, \ldots, x_{482}]$, where $\mathbf{x} \in {\mathbf{L}, \mathbf{N}, \mathbf{Y}_{\mathbf{L}}, \mathbf{u}, \mathbf{f}}$. The subscript 111 therefore indexes the first, 112 the second, 121 the third, and 482 the 64*th* element of each vector.

Thus, we can write the partial derivative of wages with respect to employment as

$$\frac{\partial \nu^{-1} \mathbf{Y}_{\mathbf{L}}}{\partial \mathbf{L}} = \nu^{-1} \begin{bmatrix} \frac{\partial Y_{L111}}{\partial L_{111}} & \cdots & \frac{\partial Y_{L111}}{\partial L_{ijk}} & \cdots & \frac{\partial Y_{L111}}{\partial L_{482}} \\ \vdots & \vdots & \vdots & \vdots \\ \frac{\partial Y_{Lijk}}{\partial L_{111}} & \cdots & \frac{\partial Y_{Lijk}}{\partial L_{ijk}} & \cdots & \frac{\partial Y_{Lijk}}{\partial L_{482}} \\ \vdots & \vdots & \vdots & \vdots \\ \frac{\partial Y_{L482}}{\partial L_{111}} & \cdots & \frac{\partial Y_{L482}}{\partial L_{ijk}} & \cdots & \frac{\partial Y_{L482}}{\partial L_{482}} \end{bmatrix}.$$
(A.1)

Due to the nested structure of the production function we have four types of partial derivatives in equation (A.1):

$$\begin{split} \frac{\partial \nu^{-1} Y_{Lqjk}}{\partial L_{qjk}} &= \frac{w_{qjk}}{L_{qjk}} \left[s_{qjk} \left\{ \frac{1}{\delta} + I^* - \frac{1}{s_q} \left(\frac{1}{\delta} - \frac{1}{\rho} \right) - \frac{1}{s_{qj}} \left(\frac{1}{\rho} - \frac{1}{\sigma_q} \right) \right\} - \frac{1}{\sigma_q} \right], \\ \frac{\partial \nu^{-1} Y_{Lqjk}}{\partial L_{ijk'}} &= \frac{w_{qjk}}{L_{ijk}} \left[s_{qjk'} \left\{ \frac{1}{\delta} + I^* - \frac{1}{s_q} \left(\frac{1}{\delta} - \frac{1}{\rho} \right) - \frac{1}{s_{qj}} \left(\frac{1}{\rho} - \frac{1}{\sigma_q} \right) \right\} \right], \\ \frac{\partial \nu^{-1} Y_{Lqjk}}{\partial L_{ij'm}} &= \frac{w_{qjk}}{L_{qj'm}} \left[s_{qj'm} \left\{ \frac{1}{\delta} + I^* - \frac{1}{s_q} \left(\frac{1}{\delta} - \frac{1}{\rho} \right) \right\} \right], \\ \frac{\partial \nu^{-1} Y_{Lqjk}}{\partial L_{q'nm}} &= \frac{w_{qjk}}{L_{q'nm}} \left[s_{q'xm} \left\{ \frac{1}{\delta} + I^* \right\} \right], \end{split}$$

where $k \neq k'$, $j \neq j'$ and $q \neq q'$, and s_{qjk} , s_{qj} and s_q denote the share of wages paid to workers in the respective cell of the labor market in the total wage bill. The index function I^* is

$$I^* = \begin{cases} \alpha - 1 & \text{in the short run,} \\ 0 & \text{in the long run,} \end{cases}$$

which follows from the production function if physical capital is fixed in the short run, i.e., if $K = \overline{K}$, and if the capital-output ratio is constant in the long-run, i.e, if $\kappa = \overline{\kappa}$.

Using the wage-setting equation in (3) we can write

$$\frac{\partial \mathbf{f}}{\partial \mathbf{u}} \frac{\partial \mathbf{u}}{\partial \mathbf{L}} = \begin{bmatrix}
\frac{\partial \phi_{111}}{\partial u_{111}} \frac{\partial u_{111}}{\partial L_{111}} & \cdots & 0 & \cdots & 0 \\
\vdots & \ddots & & \vdots \\
0 & & \frac{\partial \phi_{ijk}}{\partial u_{ijk}} \frac{\partial u_{ijk}}{\partial L_{ijk}} & & 0 \\
\vdots & & & \ddots & \vdots \\
0 & & \cdots & 0 & & \cdots & \frac{\partial \phi_{482}}{\partial u_{482}} \frac{\partial u_{482}}{\partial L_{482}}
\end{bmatrix}, \quad (A.2)$$

and

$$\frac{\partial \mathbf{f}}{\partial \mathbf{u}} \frac{\partial \mathbf{u}}{\partial \mathbf{N}} \frac{d \mathbf{N}}{d M} = \begin{bmatrix} \frac{\partial \phi_{111}}{\partial u_{111}} \frac{\partial u_{111}}{\partial N_{111}} \frac{d N_{111}}{d M} \\ \vdots \\ \frac{\partial \phi_{ijk}}{\partial u_{ijk}} \frac{\partial u_{ijk}}{\partial N_{ijk}} \frac{d N_{ijk}}{d M} \\ \vdots \\ \frac{\partial \phi_{482}}{\partial u_{482}} \frac{\partial u_{482}}{\partial N_{482}} \frac{d N_{482}}{d M} \end{bmatrix}.$$
(A.3)

Substituting the matrices (A.1), (A.2), and (A.3) into equation (6) yields the marginal employment response to immigration for the two cases of a fixed capital stock or a constant capital output ratio.

B List of variables

- *GDP*: Real GDP (West Germany) at constant 2000 prices. Source: Sachverständigenrat (2009).
- Industrial Production: Industrial production incl. construction, volume index (base year: 2000), West Germany. Source: Datastream (code BDIPTOT.G).¹⁶
- *CPI:* Consumer price index, (base year: 2000), Source: Datastream (code BDCONPRCF).
- Export performance: Share of exports in GDP. Source: OECD (2009).
- *Oil price:* Domestic crude oil price, index (base year: 2000). Source: Sachverständigenrat (2009).

¹⁶Access to datasets provided by Datastream was granted through Århus University.

• Industry mix: The variable is constructed as

$$indumix_{qj} = \sum_{h=1}^{n} g_{ht} \frac{L_{qjh,t-1}}{L_{qj,t-1}} - g_t,$$

where g_{ht} is the employment growth rate in industry h in year t, $L_{qjh,t-1}$ is the employment of education-experience group qj in industry h in year t-1, $L_{qj,t-1}$ is the aggregate employment of education-experience group qj in year t-1, and g_t the average overall employment growth rate in year t. The summation is over all two digit, non-agricultural, private-sector industries. Source: authors' calculations based on IABS.

- *Export demand*: Log of GDP per capita at constant prices and exchange rates of all OECD countries weighted by their average share in German exports during the sample period. Source: authors' calculations based on OECD (2009).
- Potential immigration: Estimated immigration potential in each education-experience group. Explanatory variables are GDP per capita at PPP, the unemployment rate and bilateral fixed effects of 20 sending countries. Source: authors' estimates based on immigration stock data from IABS and explanatory variables from the World Bank (2009) and OECD (2009).
- Average unemployment benefit: Average unemployment benefit of an unemployed of educational level q, experience group j, and national origin k. Source: IABS.
- *Ideology index:* Share of left- and right-wing parties in government weighted by their seats in parliament. Source: Bjørnskov (2008).

	ive una jo	Teryn iuoo	i juice by	cuacation	<i>group</i> , 19	00-2004			
					average	change			
	1980	1990	2000	2004	1980-04	1980-04			
		nativ	es in % of	f total		in~%			
no vocational	22	14	9	8	14	-63			
vocational	71	75	72	71	73	-2			
high school	2	4	7	9	5	440			
university	5	7	11	12	8	157			
		in persons							
total	$296,\!854$	$325,\!412$	$305,\!229$	$293,\!482$	310,413	-1			
		foreign	ters in $\%$	of total		in~%			
no vocational	62	48	37	33	45	-26			
vocational	34	46	54	56	48	134			
l =	1	2	4	5	3	594			
l	3	4	5	6	4	164			
a 1980 1980 1980 2000 2006		in~%							
total	$33,\!675$	36,436	48,312	47,162	39,740	40			

Table 1: Native and foreign labor force by education group, 1980-2004

Source: Authors' calculations based on IABS.

Figure 1: Share of foreign labor force

	<u>ln w</u>										
	$m \omega_0$	$q_{jk,t-1}$		short run long run							
oducation	cooff	50		cooff	Short run		cooff	So So		\mathbf{P}^2	obs
education	coen.	se		coen.	se		coen.	se		п	obs.
all experience groups											
$\operatorname{all}^{a)}$	0.522	(0.038)	***	-0.076	(0.007)	***	-0.158	(0.017)	***	0.99	768
no vocational	0.587	(0.059)	***	-0.063	(0.008)	***	-0.152	(0.026)	***	0.99	192
vocational	0.593	(0.060)	***	-0.077	(0.009)	***	-0.188	(0.028)	***	0.99	192
high school	0.492	(0.072)	***	-0.083	(0.021)	***	-0.164	(0.046)	***	0.99	192
university	0.445	(0.080)	***	-0.091	(0.015)	***	-0.164	(0.027)	***	0.99	192
	00	(0.000)		0.00-	(0.010)		0.202	(0.01.)		0.00	
	\exp	erience gr	oup 1	(0 - 5 ye)	ears) and e	xperie	ence grou	p 2 (6 – 1	0 year	rs)	
no vocational	0.609	(0.083)	***	-0.043	(0.009)	***	-0.110	(0.040)	***	0.99	48
vocational	0.824	(0.058)	***	-0.060	(0.008)	***	-0.341	(0.102)	***	0.99	48
high school	0.819	(0.088)	***	-0.050	(0.0135)	***	-0.275	(0.136)	**	0.99	48
university	0.434	(0.166)	***	-0.123	(0.020)	***	-0.218	(0.054)	***	0.98	48
	expe	rience gro	up 3 ((11 - 15 y	vear) and ex	xperie	nce grou	р4 (16 –	20 yea	ars)	
no vocational	0.201	(0.100)	**	-0.111	(0.020)	***	-0.139	(0.020)	***	0.97	48
vocational	0.235	(0.105)	**	-0.092	(0.015)	***	-0.120	(0.017)	***	0.98	48
high school	-0.281	(0.440)		-0.206	(0.092)	**	-0.161	(0.043)	***	0.79	48
university	0.585	(0.138)	***	-0.115	(0.036)	***	-0.278	(0.107)	***	0.87	48
	exper	rience grou	ар 5 (21 - 25 y	ears) and e	xperie	ence grou	р 6 (26 –	30 ye	ars)	
no vocational	0.435	(0.115)	***	-0.066	(0.0139)	***	-0.116	(0.025)	***	0.97	48
vocational	0.442	(0.217)	**	-0.091	(0.027)	***	-0.164	(0.070)	**	0.86	48
high school	0.658	(0.142)	***	-0.081	(0.033)	**	-0.237	(0.117)	**	0.77	48
university	0.537	(0.130)	***	-0.040	(0.019)	**	-0.087	(0.036)	**	0.85	48
	$\exp($	erience gro	oup 7	(31 - 35	years) and	exper	ience gro	up 8 (> 3	85 yea	rs)	
no vocational	0.364	(0.108)	***	-0.095	(0.022)	***	-0.149	(0.028)	**	0.98	48
vocational	0.228	(0.115)	**	-0.067	(0.016)	***	-0.087	(0.021)	***	0.98	48
high school	0.413	(0.176)	**	-0.087	(0.048)	*	-0.148	(0.109)		0.65	48
university	0.050	(0.194)		-0.067	(0.028)	**	-0.070	(0.023)	***	0.91	48

Table 2: The wage-setting curve: IV-estimation results

Notes: Dependent variable is $\ln w_{qjt}$, i.e. the log wage in each education-experience group.- Errors are heteroskedasticity robust and clustered by education-experience.- ***, **, * denote the 1%-, 5%-, and 10%-significance levels.- The model is estimated by 2SLS with group specific fixed effects.- a) A test of overidentifying restrictions cannot reject the null hypothesis that the instruments are valid (Hansen *J*-statistic, *p*-value = 0,61). The Kleibergen-Paap LM statistics rejects the null of underidentification (*p*-value = 0.00) and the Kleibergen-Paap rk Wald F statistic ($F = 83^{***}$) the null that instruments are weak.

Table 3: Partial substitutions elasticities between natives and foreigners, σ , across education-experience cells, ρ , and across education cells, δ

parameter	dependent variable	coefficient	se		observations
σ_{all}	$\ln L_{qjht}/L_{qjft}$	7.01	(0.883)	***	704
σ_{ed1}	$\ln L_{1jht}/L_{1jft}$	3.31	(0.874)	***	704
σ_{ed2}	$\ln L_{2jht}/L_{2jft}$	17.88	(2.414)	***	704
σ_{ed3}	$\ln L_{3jht}/L_{3jft}$	12.96	(5.303)	**	704
σ_{ed4}	$\ln L_{4jht}/L_{4jft}$	2.89	(0.873)	***	704
ho	$\ln L_{qjt}$	8.57	(1.448)	***	704
δ	$\ln L_{qt}$	2.86	(0.633)	***	84

Notes: Errors are heteroskedasticity robust and clustered by education-experience.- ****, **, * denote the 1%-, 5%-, and 10%-significance levels.- The equations are estimated by 2SLS.- Observations are weighted by lnL_{qjt} and lnL_{qt} .- The *F*-test rejects the null hypothesis that all coefficients σ_q are identical across educational groups (F = 29, p-value = 0.00).- A test of overidentifying restrictions cannot reject the null hypothesis that the instruments are valid (Hansen *J*-statistic, σ_{all} : *p*-value = 0.13, ρ : *p*-value = 0.18; δ : *p*-value = 0.36).- The Kleibergen-Paap *LM* statistics rejects the null of underidentification (σ_{all} : *p*-value = 0.00, ρ : *p*-value = 0.00, δ : *p*-value = 0.00) and the Kleibergen-Paap rk Wald *F* statistic (σ_{all} : $F = 56^{***}$, ρ : $F = 21^{***}$, δ : $F = 9^*$) the null that instruments are weak.

	1% increase of labor force									
		(annua	al average)							
	imperfect labor market perfect labor market							imperf. labor mark.		
	short	t-term	long	-term	short-term	long-term	lon	ig-term		
	wage	wage u-rate wage u-rat		u-rate	wage wag		wage	u-rate		

Table 4: Simulated wage and employment effects of immigration

	total labor force								
all	-0.18	0.31	0.00	0.08	-0.27	0.00	0.00	0.00	
no vocational	-0.50	1.01	-0.43	0.52	-0.89	-0.62	0.12	-0.12	
vocational	-0.13	0.16	0.06	-0.02	-0.19	0.09	-0.01	0.02	
high school	-0.12	0.14	0.08	-0.01	-0.16	0.11	-0.08	0.05	
university	-0.11	0.15	0.09	0.01	-0.16	0.11	-0.02	0.03	
				n	ative labor force				
all	-0.08	0.09	0.11	-0.06	-0.13	0.15	0.01	0.00	
no vocational	-0.03	0.13	0.12	-0.11	-0.12	0.15	0.06	-0.03	
vocational	-0.10	0.10	0.09	-0.05	-0.15	0.12	0.00	0.00	
high school	-0.08	0.07	0.12	-0.04	-0.12	0.15	-0.05	0.02	
university	0.01	0.00	0.23	-0.08	0.01	0.29	0.05	-0.02	
				fo	reign labor force				
all	-1.11	1.97	-1.09	1.16	-1.70	-1.42	-0.09	0.01	
no vocational	-1.62	3.06	-1.76	2.00	-2.74	-2.47	0.25	-0.32	
vocational	-0.54	0.94	-0.38	0.39	-0.65	-0.38	-0.18	0.22	
high school	-0.67	1.08	-0.52	0.40	-0.81	-0.53	-0.61	0.48	
university	-2.05	2.50	-2.21	1.44	-3.03	-2.76	-1.16	0.80	

wages: change in %, unemployment rate: change in %-points

Notes: The imperfect labor market scenario is based on our estimates of the wage-setting curves, the perfect labor market scenario assumes that the elasticity of the wage-setting curve approaches infinity and unemployment is zero. The short-term simulations are based on the short-run elasticities of the wage-setting curve and assume that the capital stock remains fixed. The long-run results are based on the long-run elasticities of the wage-setting curve and assume a constant capital-output ratio. In the 1%-scenario, the education and experience composition of the labor supply shock has been taken from the average distribution of the foreign labor force across the education-experience cells during the sample period. The 1980-2004 simulation is based on actual changes of the foreign labor force in each education-experience cell in each year. The annual average is calculated by weighting in each education-experience cell annual changes of wages and of the unemployment rate, respectively, with its share of in the total wage bill and in the labor force, respectively, in each year. Aggregate wage figures are calculated by weighting the wage change of each group by its share in the total wage bill. Aggregate unemployment figures are obtained by weighting each cell with its share in the labor force.

Table	B1:	Sample	selection	(W	estern	Germany,	1980) - 200.	4)
				all		native	s	foreign	ers
				obs.	in $\%$	obs.	in $\%$	obs.	in $\%$
all spe	ells		11,769	,882					
- missing nationality			1	,661					
spells with known nationality			ty 11,768	,221	100.0	$10,\!487,\!750$	100.0	1,280,471	100
- part	time wo	rkers / traine	2,543	,490	21.6	2,337,020	22.3	206,470	16.1
- age (below 15 and above 60)			60) 166	,070	1.4	152,386	1.5	$13,\!684$	1.1
- missi	ing educa	ation	183	,075	1.6	$128,\!945$	1.2	$54,\!130$	4.2
- wages below social security			y 121	,748	1.0	109,071	1.0	$12,\!677$	1.0
contril	bution ce	eiling							
total			8.753	.838	74.4	7.760.328	74.0	993.510	77.6

Notes: The IABS includes only wage and salary workers but no self-employed. Due to changes in methodology, allocation of Berlin to the newly formed German states since 1999, and the German unification aggregate statistics for the entire observation period and the region used is not available. During 2000-2004 on average 90 percent of the entire German workforce were dependent employees and 81 percent worked in Western Germany. Thus, about 73 percent of the entire German workforce is covered by our analysis.

Source: Authors' calculations based on the IABS, Statistisches Bundesamt, GENESIS-Online.